Author's response to reviews

Title: Response of lymphocyte subsets and cytokines to Shenyang prescription in Sprague-Dawley rats with tongue squamous cell carcinomas induced by 4NQO

Authors:

Canhua Jiang (canhua-j@sohu.com)
Dongxia Ye (yedongxia122@hotmail.com)
Weiliu Qiu (gwl@omschina.org.cn)
Xiuli Zhang (zhxl59@yahoo.com.cn)
Zhiyuan Zhang (zhzy@omschina.org.cn)
Di He (enamel1981@hotmail.com)
Ping Zhang (pingzhang73@hotmail.com)
Wantao Chen (chenwantao2002@hotmail.com)

Version: 3 Date: 30 November 2006

Author's response to reviews:

Dec.1, 2006
Dear Iratxe Puebla,

Thank you for your letter of 11/24/06 indicating our paper Response of lymphocyte subsets and cytokines to Shenyang prescription in Sprague-Dawley rats with tongue squamous cell carcinomas induced by 4NQO (MS: 1728714112121869) by Jiang et al can be resubmitted with revision. I appreciate the prompt and constructive review. Dr. Angela Panther at the University of Aberdeensee the letter from pat115@abdn.ac.uk has helped to revise the manuscript as suggested by reviewers.

Specific points include the following:

Major:
Particular attention has been paid to the abstract. Its Background section was revised carefully.

Minor:
Language revisions was carried out by a commercial copyediting service.

I believe that the suggested changes have considerably strengthened the current version. I greatly appreciate your consideration of our work again.

Sincerely,

Wantao Chen
Professor, Department of Oral and Maxillofacial Surgery, Ninth People's Hospital, School of Stomatology, Shanghai Jiao Tong University
Vice Director, Shanghai Key Laboratory of Stomatology, 639 Zhizaoju Road, Shanghai, P.R. China, Tel:86-21-63138341-5211, Fax:86-21-63135412, E-mail: chenwantao2002@hotmail.com